Author’s response to reviews

Title: Hypoxia and hypotension in patients intubated by physician staffed helicopter emergency medical services - a prospective observational multi-centre study

Authors:
Geir Arne Sunde (geir.arne.sunde@norskluftambulanse.no)
Mårten Sandberg (marten.sandberg@medisin.uio.no)
Richard Lyon (richardlyon@doctors.org.uk)
Knut Fredriksen (Knut.Fredriksen@unn.no)
Brian Burns (bburns_2000@yahoo.com)
Karl Ove Hufthammer (karl.ove.hufthammer@helse-bergen.no)
Jo Røislien (jo.roislien@norskluftambulanse.no)
Akos Soti (akossoti@gmail.com)
Helena Jantti (Helena.Jantti@kuh.fi)
David Lockey (David.Lockey@nbt.nhs.uk)
Jon-Kenneth Heltne (jkhe@helse-bergen.no)
Stephen Sollid (stephen.sollid@norskluftambulanse.no)

Version: 2 Date: 21 Jun 2017

Author’s response to reviews:

Dear Editors

We apologise for not recognising that the date of trial registration was missing, and have now amended this at the end of the abstract.
Yours sincerely

Geir Arne Sunde

For the authors